Author's response to reviews

Title: Validating MOSPA questionnaire for measuring physical activity in Pakistani women

Authors:

Romaina Iqbal (romaina@ccc.mcmaster.ca)
Ghazala Rafique (ghazala.rafique@aku.edu)
Salma Badruddin (salma.badrudinn@aku.edu)
Rahat Qureshi (rahat.qureshi@aku.edu)
Katherine Gray-Donald (katherine.gray-donald@mcgill.ca)

Version: 2 Date: 25 April 2006

Author's response to reviews: see over
To
Dr. Hiromichi Kumagai
Deputy Editor-in-Chief
Nutrition Journal

Re: Validating MOSPA questionnaire for measuring physical activity in Pakistani women - Manuscript ID 1216085962935737

Dear Dr. Kumagai,

I would like to thank you for considering our manuscript for publication in the Nutrition Journal. We found all the concerns raised by the reviewers highly relevant and have made all the necessary corrections in our manuscript. I hope you will find our revisions satisfactory.

I look forward to hearing from your office soon.

Sincerely
Katherine

Katherine Gray-Donald Ph.D.
Associate Professor
School of Dietetics and Human Nutrition
McGill University, Canada
21, 111 Lakeshore Road
Ste-Anne-de-Bellevue
H9X 3V9 QC Canada
Ph: 514 398 7677
Fax 514 398 7739